

Chasing the ghost of fracking in the Vaca Muerta Formation: Induced seismicity in the Neuquén Basin, Argentina

Ryan Schultz 💿 *1, Guillermo Tamburini-Beliveau 💿², Sebastián Correa-Otto 💿³, Javier A. Grosso-Heredia^{4,5}

¹Swiss Seismological Service, ETH Zürich, Zürich, Switzerland, ²Consejo Nacional de Investigaciones Científicas y Técnicas – Centro de Investigaciones y Transferencia de Santa Cruz, Río Gallegos, Argentina, ³Consejo Nacional de Investigaciones Científicas y Técnicas, Buenos Aires, Argentina, ⁴Departamento de Geografía, Universidad del Comahue, Neuquén, Argentina, ⁵Instituto de Formación Docente Fiske Menuco, General Roca. Río Negro

Author contributions: Conceptualization: R. Schultz. Data Curation: R. Schultz, G. Tamburini-Beliveau, S. Correa-Otto. Formal Analysis: R. Schultz. Funding Acquisition: . Investigation: R. Schultz. Methodology: R. Schultz. Project Administration: R. Schultz. Resources: R. Schultz. Software: R. Schultz. Supervision: R. Schultz. Validation: R. Schultz, G. Tamburini-Beliveau, S. Correa-Otto. Visualization: R. Schultz. Writing – original draft: R. Schultz, G. Tamburini-Beliveau, S. Correa-Otto. Writing – review & editing: R. Schultz, G. Tamburini-Beliveau, S. Correa-Otto, J.A. Grosso-Heredia.

Abstract Earthquakes are known to be induced by a variety of anthropogenic causes, such as hydraulic fracturing. In the Neuquén Basin of Argentina, hydraulic fracturing has been used to produce hydrocarbons trapped in the shales of the Vaca Muerta Formation. Correspondingly, incidences of seismicity there have increased. We collect information on well stimulations and earthquakes to perform statistical analysis linking these two datasets together. Spatiotemporal association filters suggest that the catalogue of events is biased towards hydraulic fracturing operations. After accounting for false-positives, we estimate that ~0.5% of operations are associated with earthquakes. These associated event-operation pairs show highly correlated temporal signals (>99.99% confidence) between seismicity/injection rates. Based on this evidence, we argue that many of these earthquakes (up to M4) are induced. We support this argument by comparing the geological setting of the Neuquén Basin against conditions needed for fault reactivation in other susceptible/seismogenic basins. This recognition adds to the growing list of (hydraulic fracturing) induced seismicity.

Non-technical summary Earthquakes have been increasingly encountered in Argentina's Neuquén Basin, a previously seismically quiet region. Simultaneously, hydraulic fracturing operations have been targeting shales in the Vaca Muerta Formation. Hydraulic fracturing injects subsurface fluids under high enough pressure to split rocks, allowing access to the hydrocarbons within. This type of operation is known to cause earthquakes elsewhere. To determine if these earthquakes were caused by hydraulic fracturing, we performed statistical tests. To high-degrees of confidence, we find that many events tend to be clustered around hydraulic fracturing wells in time and space. Sensitivity testing suggests that these correlations almost certainly did not occur by accident. Faults, stress information, and other factors in the Neuquén Basin are comparable against other basins that have already induced earthquakes. We conclude that many of the events here were induced. We discuss how future research could better answer some of our unresolved questions.

1 Introduction

Induced seismicity caused by subsurface injection operations, like wastewater disposal, has been well documented since the 1960s (Healy et al., 1968; Raleigh et al., 1976). Since 2010, earthquakes in North America have proliferated, following the advent of hydraulic fracturing for shale gas (Foulger et al., 2018). Within the United States, these earthquakes were predominantly caused by the increased disposal of frack-related produced water (Ellsworth, 2013). However, earthquakes have also been induced directly by hydraulic fracturing operations around the world (Schultz et al., 2020; Atkinson et al., 2020). For example, seismicity in the Horn River Basin (Farahbod et al., 2015), Duvernay Formation (Bao and Eaton, 2016; Schultz et al., 2017), Montney Formation (Mahani et al., 2017; Peña Castro et al., 2020), Bowland Shale (Clarke et al., 2019; Kettlety et al., 2020), Appalachian Basin (Skoumal et al., 2015; Brudzinski and Kozłowska, 2019), Delaware Basin (Skoumal et al., 2020), and Sichuan Basin (Lei et al., 2017, 2019) has been linked to hydraulic fracturing.

Typically, these earthquakes occur when the growth of stimulated fractures intersect pre-existing faults (Galloway et al., 2018), reactivating susceptible portions of the fault via a reduction in effective normal stress (Moein et al., 2023). One of the largest documented cases (M_L 5.7) occurred in the Sichuan Basin of China, causing noteworthy damage and injuries (Lei et al., 2019). Many cases have encountered felt ground shaking, or events as large as M4 (Bao and Eaton, 2016;

Froduction Editor: Yen Joe Tan Handling Editor: Giuseppe Petrillo Copy & Layout Editor: Hannah F. Mark

Received: August 26, 2024 Accepted: November 12, 2024 Published: November 27, 2024

^{*}Corresponding author: Ryan.Schultz@sed.ethz.ch



Figure 1 Comparison between seismotectonic setting and hydraulic fracturing operations. (a) Map showing part of Argentina, alongside boundaries of nearby shale basins (red polygons). Inset polygon shows the location of the Neuquén Basin within South America. (b) Map of the study area shows earthquakes (red circles) and hydraulic fracturing operations (blue diamonds), alongside the city of Neuquén (white circle, with text). Faults from prior studies (Silvestro and Zubiri, 2008; Cristallini et al., 2009) are also shown (white lines).

Schultz et al., 2017; Mahani et al., 2017). In response, some jurisdictions have emplaced moratoriums on resource development because of induced earthquake concerns (Kettlety et al., 2020). Typically, these risks are managed through a traffic light protocol (Bommer et al., 2006; Häring et al., 2008; Ader et al., 2019), which designates thresholds for operations to begin mitigation (yellow-light) and a regulatory interventions (redlight). Traffic light protocols have been implemented around the world to manage hydraulic fracturing induced earthquakes, using various red-light thresholds. New approaches have begun to define these red-light thresholds using risk-based principles (Schultz et al., 2021a,b, 2023a).

Within Argentina, there are several basins which have been appraised for or targeted by hydraulic fracturing (Figure 1), the most productive of which has been the Neuquén Basin (US-EIA, 2015). The Neuquén Basin (~160,000 km²) is situated in central-eastern Argentina and bounded by the Andes to the west (Urien and Zambrano, 1994; Marchal et al., 2020). For comparison, the Duvernay Formation covers an area of ~130,000 km² (US-EIA, 2013). The Neuquén Basin contains Late Triassic to Early Cenozoic strata that were deposited in a back-arc tectonic setting, including accumulated deposits of siliciclastics, carbonates, and evaporites (Howell et al., 2005). Most relevant to hydraulic fracturing development have been the Vaca Muerta Formation and the Los Molles Formation (Kugler, 1985; Ramos et al., 2020). The Vaca Muerta Formation is Late Jurassic -Early Cretaceous aged shale consisting of marls, organic shales, and lime-mudstones; in some regions, thicknesses can reach up to 520 m and depths of 3,150 m (Badessich et al., 2016; Legarreta and Villar, 2015). The deeper Los Molles Formation is Middle Jurassic aged shale consisting of organic shales and lime-mudstones (Martínez et al., 2008). The crystalline basement consists of Silurian-Devonian igneous-metamorphic rocks, intruded by magmatic arc granitoids (Cingolani et al., 2011). Initial estimates of technically recoverable natural gas were 308 Tcf and 275 Tcf for the Vaca Muerta Formation and Los Molles Formation, respectively (US-EIA, 2013). Initial estimates of technically recoverable oil were 16.2×10⁹ bbl and 3.7×10⁹ bbl for the Vaca Muerta Formation and Los Molles Formation, respectively (US-EIA, 2013). The Neuquén Basin is host to one of the most significant sources of unconventional hydrocarbons in the world and is an important energy source for Argentina (Minisini et al., 2020). Currently, the stateowned oil and gas company YPF holds the largest share of unconventional investments in the Neuquén Basin (Lu et al., 2024).

Like many hydraulic fracturing basins, the seismic history of the Neuquén Basin is quiescent and sparsely recorded. Past studies have understandably focused on the constraining the hazardous seismicity from the Nazca Plate subduction (Bilek, 2010). In some cases, temporary/peripheral surveys may have captured nearby seismicity (Bohm et al., 2002). Inferences from structural geology suggest a complex history, including extensional and compressional periods (Marchal et al., 2020). Recent (Cenozoic) tectonism is mostly related to compression from the Andean deformation (Cobbold and Rossello, 2003). Tectonism and faulted structures are thought to have played a role in the migration of hydrocarbons through the geological history of the Neuquén Basin (Mosquera et al., 2008). Due to concerns of induced seismicity (Cunningham, 2022;



Figure 2 Earthquake magnitude statistics. Magnitude-frequency distribution of events in the regional catalogue: data counts are plotted as both the cumulative (circles) and non-cumulative (bars) distributions, alongside the best fit to the data (solid line) and the magnitude-of-completeness (dashed lines). Relevant parameters are reported as text.

Surma, 2024), recent work has begun to constrain the regional seismicity of the Neuquén Basin (Curia et al., 2018; Correa-Otto, 2021; Correa-Otto et al., 2024). In fact, it has been suggested that seismicity in the Neuquén Basin is induced (Tamburini-Beliveau et al., 2022).

Delving deeper into the socio-environmental aspects, these recent earthquakes have impacted the region in several ways. In the rural community of Sauzal Bonito, earthquake ground shaking is regularly felt, and homes have sustained damage (Surma, 2024). Consequently, in 2022, the state committed to the construction of 56 new basic earthquake-resistant homes (12 delivered to date) which were 50% funded by one of the operating companies (LM Neuquen, 2024). Additionally, the region's cliffsides have experienced multiple landslides, resulting in fatalities and injuries (Álvarez Mullally, 2021; Nacion, 2022). Industrial operations have not been immune to the seismic activity; some operators have temporarily suspended hydraulic fracturing activities due to the associated risks of induced seismicity (Beascochea, 2020). While this socio-environmental information is limited (and primarily based on testimonies and news reports), it provides motivation for further scientific inquiry.

Within this study, we begin to the discern causal factors for seismicity in the Neuquén Basin. To do so, we compile information regarding hydraulic fracturing stimulations and regional catalogues of earthquakes. We then use spatiotemporal association filters to link candidate operations with sequences of earthquakes. Furthermore, cross-correlation reshuf-fling tests are used to determine the statistical confidence of well-earthquake associations. We find significant evidence (>99.99% confidence) for a correlation between hydraulic fracturing operations and many of the strongly-associated earthquakes. We then highlight the relevant geological/operational factors relevant for causing induced earthquakes, in comparison with other known cases. Last, the limitations of this analysis/dataset are also discussed – to highlight future research directions to better understand these earthquakes and further disentangle their causes.

2 Input Datasets

2.1 Earthquake catalogue

Earthquake catalogue information is compiled from multiple sources. Predominantly, these sources include the National Institute of Seismic Prevention of Argentina (Instituto Nacional de Prevención Sísmica, INPRES), a PhD project cataloguing earthquakes in the basin (Correa-Otto, 2021; Correa-Otto et al., 2024), and miscellaneous prior studies (Vásquez et al., 2020; Tamburini-Beliveau et al., 2022). The INPRES catalogue spans from 2015 onwards (predominantly starting in 2019), while the PhD project catalogue predominantly spans from 2014-2020.

The latter (PhD) catalogue was obtained by a local broadband seismological network that was established in the central-northern region of the Neuquén province (Figure S1). It comprised 11 stations covering an area of approximately 70×70 km², which consistently collected data from November 2014 to July 2016. From July 2016 until June 2020, the number of stations operational in the network was reduced to 5. Each station was outfitted with a Trillium 120PA broadband seismometer and a Taurus digital recorder (Correa-Otto, 2021). This catalogue predominantly comprises the earlier segment of our study period.

The INPRES catalogue spans from 2015 onwards and constitutes data requests from INPRES. Limited station availability hampered event detection (~10 events from



Figure 3 Schematic diagram defining the SAF spatial/temporal-scores. (a) The spatial-score of a single event-well pair decaying as a function of distance (lines, see legend). (b) The temporal-score of a single event-well pair decaying as a function of time (lines, see legend). Intervals separating the bi-linear pieces are denoted by vertical dashed lines.

2015-2019). Partly because of network expansion in 2019 (Figure S1), this catalogue starts detecting events more routinely (~360 events from 2019 onward). This catalogue predominantly comprises the later segment of our study period.

We collated these catalogues together into a single combined catalogue for this study. To ensure the uniqueness of events, we identify those which are contemporaneous and then merge them. The preferred origin is selected as the one from the PhD project catalogue. The limited overlap between catalogues means very few events are flagged as identical. Our combined catalogue starts in November 2014 and spans until March 2024 (Figure S2). There are 545 events in this catalogue. Event depths range from the surface to 66 km; with a mean value of ~7 km and a standard deviation of ~6 km. The largest event in our catalogue compilation is M 5.0 and occurred on 7 March 2019. There have been more than 10 M4+ events documented during the recording interval. A simple fit to the magnitudefrequency distribution indicates a *b*-value of 0.79±0.04 (Figure 2). The magnitude-of-completeness appears at M ~2.5, which is the value asserted by INPRES.

2.2 Hydraulic fracturing information

The data for wells location, injection and other characteristics were downloaded in plain text format from the official public databases from the Argentina Energy Secretariat and the Neuquén province's Ministry for Energy and Natural Resources. The databases were combined and cleaned to avoid duplicated registries and other inconsistencies (Figure S3). Data on hydraulic fracturing includes surface well locations, stimulation start/end times, total fluid injected, number of stages, and other information less relevant for this study. Unfortunately, the start/end times of 16% of hydraulic fracturing wells (Figure S4) are missing in the original source files. We are unable to assert if or how many wells are enitrely missing from this database. These missing cases will evade scrutiny in our subsequent analysis, since both spatial and temporal information are needed.

The economic potential of the Vaca Muerta Formation, as a hydraulic fracturing target, was first recognized in 2010 by Repsol-YPF. During this early phase (2010-2012), wells tended to inject modest amounts of fluid during stimulation programs $(10^{1}-10^{3} \text{ m}^{3})$. However, similar to other hydraulic fracturing basins, the scale, complexity, and frequency of operations increased with time. By 2019, the vast majority of operations are injecting $10^{4}-10^{5} \text{ m}^{3}$, per well. To date, there have been more than 4,000 unconventional wells stimulated in the Neuquén Basin.

3 Methods & Results

3.1 Screening for wells via a spatiotemporal association filter (SAF)

The first step in determining the propensity for induced earthquakes is to screen for events that are close to operations in time and space. This rationale is based on standard criteria for identifying a causal induced seismicity relationship, which prioritizes the examination of relationships in time-and-space (Davis and Frohlich, 1993; Verdon et al., 2019; Foulger et al., 2023). In this study, we use a spatiotemporal association filter (SAF) to screen for induced events. Comparable approaches have been used for identifying induced seismicity caused by hydraulic fracturing (at the basin-scale) in the past (Atkinson et al., 2016; Schultz et al., 2018; Lomax and Savvaidis, 2019; Savvaidis et al., 2020; Ghofrani and Atkinson, 2020, 2021).

To quantify the linkage between earthquakes and operations, we start by defining a simple bi-linear decay function: $L(\mathbf{X}; X_1, X_2, P_1)$. Over the interval 0- X_1 function scores decline from 1- P_1 , and over the interval X_1 - X_2 function scores decline from P_1 -0. For any candidate event-well pair (i, j), the spatial-score $L_{i,j}(\mathbf{r}; r_1, r_2, P_1)$ and a temporal-score



Figure 4 Spatiotemporal association filtering. (a & c) Maps with locations of earthquakes (circles) and hydraulic fracturing operations (diamonds). (b & d) Temporal timings of hydraulic fracturing operations (two diamonds linked with a blue line) and earthquake magnitudes (circles). Comparisons are made between the full (top row) and filtered (bottom row) datasets. Unassociated operations/earthquakes have faded colors.

 $L_{i,j}(\mathbf{t}; t_1, t_2, P_1)$ are assigned using this bi-linear function, with unique window intervals $(r_1 \cdot r_2, t_1 \cdot t_2)$ and score cut-off point (P_1) . Next, temporal and spatial score components are combined into a single event-well SAF-score by multiplication: *i.e.*, $L_{i,j} = L_{i,j}(\mathbf{r}) \times L_{i,j}(\mathbf{t})$. For a given operation j, SAF-scores are computed for all events and the total SAF-score is a summation of all single-event scores $SAF_j = \sum_i L_{i,j}$. Note that we also apply a uniqueness condition, so that any individual event i can only be associated with a single (highestscoring) operation j, to avoid double-counting events.

In applying this approach to the Neuquén Basin, we use our compiled catalogue of earthquakes and hydraulic fracturing database (Section 2). For the spatial component of SAF-scores $L_{i,j}(\mathbf{r})$, earthquakes that are within 7 km of an operation are scored between 1.00-0.95, with earthquakes between 7–15 km scoring 0.95-0.00. We feel that the choice of 7–15 km interval for the spatial-score is justified, based on the reported spatial resolution of catalogued earthquakes. In cases of poorer spatial resolution, SAF spatial intervals (r_1-r_2) should match this resolution, to avoid dis-

carding true associations because of this variance. If a high-resolution catalogue was available instead, intervals of 1-3 km would be more appropriate, since the furthest documented case of induced seismicity from a hydraulic fracturing well was ~1.5 km (Schultz and Wang, 2020) - although this would require also having information on the horizontal trajectory of wells, which is also unavailable to us. For the temporal-scores $L_{i,j}(\mathbf{t})$, any earthquakes that occurs during well stimulation period are assigned a score of 1.0; anything that occurs post completion decays linearly to 0.0 by 30 days afterwards. This 30-day interval is justified, based on decays of trailing seismicity typically following hydraulic fracturing in other basins (Schultz et al., 2022, 2023b). For clarity, we also provide a visual definition of how we individually assign spatial-scores $L_{i,j}(\mathbf{r})$ and temporalscores $L_{i,j}(t)$ (Figure 3). Our SAF processing choices are comparable to those used in previous studies (Lomax and Savvaidis, 2019; Savvaidis et al., 2020; Ghofrani and Atkinson, 2020, 2021).

The application of the SAF begins to identify potential induced seismicity cases (Figure 4). We consider only wells with summed SAF-scores greater than 1.0, which identifies 46 potential cases for further consideration. The case with the highest SAF-score is ~12. The two largest earthquakes associated with an operation are the M 4.3 event on 16 November 2018 and the M 4.1 event on 13 April 2022. Associated cases predominantly occur after 2019, in the region surrounding the Cerros Colorados Complex (i.e., the Los Barreales and Mari Menuco Reservoirs). Although, we do note associated cases as early as late-2014. Likely, this temporal skew has been influenced by monitoring improvements alongside increases in fracking intensity. Of the total database of unconventional wells, ~1.6% operations were associated with earthquakes and ~42% of the earthquakes were associated with hydraulic fracturing operations. However, this analysis has the potential for false-positives – spurious associations between earthquakes and operations.

We ensure the robustness of our results through bootstrap resampling. To do so, we take the catalogue of events and randomly reshuffle their origin times. This process destroys any meaningful earthquake-operation correlations. Said another way, the reshuffling process is used to estimate the rate of false-positives, given our SAF design choices. Considering the potential for falsepositives is especially important here, given the previously described unknowns/uncertainties of our input datasets (Section 2). We then take this randomly reshuffled catalogue and perform the exact same SAF processing. This is repeated 500 times, to query the variability of (false-positive) SAF scores as a statistical distribution (Figure 5). After reshuffling, the SAF processing is no longer able to produce similarly high percentages of association, suggesting an average false-positive association of ~1.1% (Figure 5a). This means there is only a 1/500 chance of producing an association as high as the 'regular' value by accident. If we next examine the distribution of SAF-scores (Figure 5b), we see that the two variants are visually distinct. We estimate that the maximum (false-positive) SAF-score is ~5.5 on average, which is much lower than the regularly observed value. Furthermore, we perform the Kolmogorov-Smirnov test (Berger and Zhou, 2014) to statistically quantify differences between the regular/reshuffled distributions of SAF-scores (Figures 5b & S5). To ~99.8% confidence, the distribution of these two SAF-score variants are significantly different.

We further assess the robustness of our results through a perturbation analysis. In this test, we modify the parameters and functional form of the spatial/temporal-sores during SAF processing, to examine the influence these subjective choices have on our results. Here, we consider five alternative SAF processing approaches. In the first alternative, the same $L_{i,j}(\mathbf{r})$ and $L_{i,j}(\mathbf{t})$ bilinear functions are used, just with more restrictive spatial-score cutoffs (*i.e.*, 5-10 km for r_1 - r_2). In the four remaining alternatives, we consider spatial/temporal-scores and functions/parameters previously used for identifying seismicity induced by hydraulic fracturing (Lomax and Savvaidis, 2019; Savvaidis et al., 2020; Ghofrani and Atkinson, 2020, 2021). Then, the SAF processing and bootstrapping is repeated for all these alternative formulations (Table 1 & Figures S6-S15). Despite the differences between the SAF alternatives, the results consistently suggest a statistically meaningful association between hydraulic fracturing and earthquakes. Overall, the results of this section are indicative of some physical process biasing the triggering of earthquakes in the Neuquén Basin.

3.2 Quantifying the strength of associations through the cross-correlation reshuffling test

To further investigate the source of this biased earthquake triggering, we next consider the crosscorrelation reshuffling test (Oprsal and Eisner, 2014; Schultz and Telesca, 2018). In this test, rates of seismicity are compared with hydraulic fracturing injection rates using cross-correlations. The cross-correlation quantifies the strength of time-lagged correlations between these two signals. The statistical significance of this time-lagged correlation can be discerned via a reshuffling of the input timeseries' spectral phase information (Schultz and Telesca, 2018).

In applying this approach to the Neuquén Basin, we first filter our earthquake catalogue above the magnitude-of-completeness (Figure 2) to account for temporally changing detection thresholds. We consider the correlation between the top ten SAF-screened operations (Section 3.1) and their associated earthquakes (Figure 6a), to ensure that only adequately identified cases are considered. Due to limited data availability, we simply consider an effective injection rate timeseries, which is the total amount of volume injected at a well divided by the total duration of stimulation start/stop times. Catalogue information is binned into a timeseries of daily earthquake rates. Reshuffled crosscorrelation tests place very high confidence (>99.99%) on these two timeseries being associated with each other (Figure 6b). Similarly, tests performed on the top SAF-scoring operations as individual sequences, also shows strong (up to ~99.7%) indications of an association (Figures S26-S31). Unfortunately, these tests are unable to meaningfully discern lag times between start of operations and responding seismicity, because of the low temporal resolution of the well database. Discrete timings of individual stage stimulations (and better catalogue detections) would be needed to improve the results here. These results are further suggestive of hydraulic fracturing playing a role in triggering some of the events in the Neuquén Basin.

4 Discussion

4.1 Comparison against known cases of hydraulic fracturing induced seismicity

Much of the Neuquén Basin seismicity is likely induced, with hydraulic fracturing being a strong candidate for the anthropogenic source. This is based on statistical testing that indicated biased spatiotemporal associations (Section 3.1; Figures 4 & 5) and statistically significant correlations (Section 3.2; Figure 6) with ongo-



Figure 5 Bootstrapped SAF results. (a) The basin-scale percentage of wells associated with earthquakes via SAF (red-line), in comparison to the association percentages when reshuffling catalogues (grey bars). (b) The distribution of SAF-scores for the regular dataset (red bars), in comparison with reshuffled data (grey bars).

Study	% Association	(FP) % Assoc	Association chance	Max SAF	(FP) Max SAF	KS-test
This study	1.58	1.11±0.13	~ 1/500	12.40	5.47±1.22	99.79%
This study (w/ restrictive spatial- score)	1.13	0.55±0.10	≪1/500	12.02	3.69±0.93	99.98%
Lomax and Savvaidis (2019)	0.65	0.11±0.05	≪1/500	9.09	2.08±0.61	97.89%
Savvaidis et al. (2020)	1.03	0.50±0.10	< 1/500	12.36	3.43±0.87	99.96%
Ghofrani and Atkinson (2020)	1.20	0.72±0.10	≪1/500	10.42	3.85±0.80	99.25%
Ghofrani and Atkinson (2021)	1.13	0.66±0.09	≪1/500	9.21	3.78±0.75	95.39%

Table 1 Summary of alternative SAF processing. Results for SAF-scores and bootstrapping results are summarized for five SAF processing alternatives. See Figure 3 for a visualization of spatial/temporal-scoring functions. FP is an abbreviation for false-positive here.

ing operations. These spatiotemporal indicators are key criteria for declaring a sequence of events as induced (Davis and Frohlich, 1993; Verdon et al., 2019; Foulger et al., 2023).

To continue building a case for causality, we also consider a comparison against known cases of hydraulic fracturing induced seismicity. Typically, these induced earthquakes are observed close to their causal operations (Eaton et al., 2018; Fasola et al., 2019; Wang et al., 2020), occurring predominantly during stimulation (Schultz et al., 2017, 2022; Clarke et al., 2019), and preferentially associated with stimulation of deep/basal shales (Skoumal et al., 2018). Of the best identified cases, we see a similar correspondence (Figures S16-S25): all earthquakes are clustered nearby to their operations with earthquakes occurring frequently during stimulation intervals and fewer trailing events.

Often this proximity is interpreted as related to the causal mechanisms: stimulated fractures need to encounter pre-existing faults that allow for hydraulic communication to susceptible portions of the fault (Galloway et al., 2018). In this sense, nearby basement-rooted faults cross-cutting the targeted shale are an important indicator for seismogenic operations (Corlett et al., 2018). Within the Neuquén Basin, the complex tectonic history has left a series of such faults, includ-

ing transtensional and transpressional flower structures (Marchal et al., 2020), which are often associated with induced earthquakes in other basins. Furthermore, these faulted structures have been implicated in fluid-flow events of the geological past (Mosquera et al., 2008), providing a plausible means for reservoirbasement pore pressure communication. Interestingly, hydrocarbon maturation has contributed to overpressure reservoir conditions for much of the Vaca Muerta Formation (Varela et al., 2020). In other basins, overpressure has been another indicator of earthquake susceptible regions (Eaton and Schultz, 2018). Overall, the geological conditions needed to encounter hydraulic fracturing induced seismicity seem to be intact for the Neuquén Basin (and consistent with other cases).

In highly susceptible sub-basin regions, the activation rate of operations can be large (up to tens of percent) (Schultz et al., 2018); however, this can vary considerably over a few kms, with only a paucity of wells inducing events at the basin-scale (Atkinson et al., 2016; Brudzinski and Kozłowska, 2019). This is often interpreted as due to the rarity of encountering all susceptibility conditions simultaneously, due to subsurface heterogeneities (Pawley et al., 2018; Hicks et al., 2021). These factors are likely relevant for the Neuquén Basin, given the strong clustering of cases within the opera-



Figure 6 Cross-correlation reshuffling tests. (a) Timeseries data of earthquake rates above the magnitude-ofcompleteness (red bars), earthquake rates for the full catalogue (grey bars), and effective injection rates (blue lines), for the top ten SAF-scoring operations. (b) cross-correlation of timeseries data (purple line) alongside confidence intervals (black lines) from reshuffling tests.

tional footprint (Figure 4). Despite these complexities, constraining this activation rate can be important for hazard analysis (Ghofrani and Atkinson, 2020, 2021; Verdon and Bommer, 2021). Our raw rate of activation in the Neuquén Basin (i.e., ~1.6%) likely overestimates the true rate, due to the potential for false-positives during SAF processing (Section 3.1). If we simply subtract off the false-positive rates, all our SAF alternative processing agree on a corrected rate of ~0.5% \pm 0.1%. We note that this value is on the same order-of-magnitude as other basins around the world (Schultz et al., 2020). On the other hand, our analysis has neglected to account for potential false-negatives – true cases of hydraulic fracturing induced seismicity that were misidentified by SAF processing. Accounting for these types of errors is nebulous, so our estimate should be considered as a lower bound. Certainly, monitoring improvements can demonstrably influence the estimation of this parameter (Ghofrani and Atkinson, 2020). Higher resolution databases would be needed to provide further interpretation.

4.2 Limitations of our analysis

While it appears clear that some of the earthquakes in the Neuquén Basin are induced by hydraulic fracturing, there still remains ambiguous events: for example, the largest earthquake within our catalogue, the M5 event on 7 March 2019. Similarly, the majority of the M4 events were not associated with an operation (Figure 4). In fact, only 42% of the catalogued earthquakes were linked with operations via our SAF. Considering these observations, we discuss the potential deficiencies and limitations of our analysis.

Part of this can be alleviated slightly through SAF optimization. For example, the M 4.2 event on 25 September 2017 occurring just outside the range of our temporal window (Figure S16). However, we suspect that this is a relatively minor factor, given the relative agreement between all SAF processing alternatives. Another issue confounding our analysis (and potentially causing an underreporting of cases) is the incompleteness of input datasets (Figure S4). Of the reported ~4000 well locations, roughly 16% are missing data critical for our analysis. Similarly, the M5 event (7 March 2019) happens one day before the reported start time of a nearby well (*i.e.*, ~2.5 km southwest from '*YPF.Nq.LLLO.x-2(h)*'). It is possible that many events could still be associated with hydraulic fracturing operations that were only poorly or coarsely documented.

On the other hand, there is the potential that these residual events could also be associated with other anthropogenic operations. Globally, induced seismicity has been observed in correspondence with numerous operation types (Foulger et al., 2018). Operations relevant for the Neuquén Basin include reservoir impoundment (Gupta, 2002), hydrocarbon production (Muntendam-Bos et al., 2022), or wastewater disposal (Ellsworth, 2013). Even some component of natural/baseline seismicity could play a role here, given the propensity for regions of seismic activity outside of the Neuquén Basin (Correa-Otto et al., 2018, 2024). The observation of multiple anthropogenic sources can complicate the identification of the true triggering mechanism (Zhai et al., 2021).

Regarding reservoir impoundment The Cerros Colorados Complex is a series of four embankment dams and a hydroelectric facility with a capacity of 462 MW (Dalmati et al., 2018), related to the Los Barreales and Mari Menuco Reservoirs along the Neuquén River (Figures 1 & 4). These two reservoirs have mean depths of ~67 m and ~80 m, respectively, and constitute a large water body covering more than 587 km². These facilities have been in operation for decades, with the facility beginning operation in 1972 and construction officially completed on 31 October 1980. To our knowledge, no reports of induced seismicity have been linked to the impoundment of these reservoirs (or any of the other

nearby Argentinian reservoirs).

Regarding hydrocarbon production Unconventional production in the Neuquén Basin is the dominant source of hydrocarbons for Argentina. Since the start of unconventional exploitation in the Neuquén Basin, rates of oil and gas production have steadily risen. By 2017, oil/gas production rates were $\sim 2 \times 10^6$ m³/year and $\sim 3 \times 10^9$ m³/year, respectively; cumulative amounts were $\sim 8 \times 10^6$ m³ and $\sim 7 \times 10^{12}$ m³, respectively (Rosa and D'Odorico, 2019). To date, no studies have linked induced seismicity with hydrocarbon production following hydraulic fracturing anywhere throughout the world; although, this could change with time given that induced seismicity has been associated with conventional hydrocarbon production (Wetmiller, 1986; Jacquemond et al., 2024; Muntendam-Bos et al., 2022).

Regarding wastewater disposal Generally, the proper management of water is a significant concern for the Neuquén Basin (Forni et al., 2018, 2021). During unconventional oil production, roughly one third of the produced fluid is wastewater, which is then disposed of via subsurface injection (only a 5% recycling fraction is reported) (Rosa and D'Odorico, 2019). Basin-scale disposal rates are negligible prior to 2012, but quickly rises to ~10⁶ m³/year by 2014 (Rosa and D'Odorico, 2019); by the end of 2020, there had been $\sim 5 \times 10^6$ m³ of wastewater disposed of in the Neuquén Basin (Tamburini-Beliveau et al., 2022). Better understanding the potential for disposal earthquakes will likely be important here; especially considering the prevalence of earthquakes caused by comparable wastewater disposal of elsewhere (Horton, 2012; Ellsworth, 2013; Frohlich et al., 2016; Zhai et al., 2021; Schultz et al., 2023c).

Ultimately, these kinds of limitations are best addressed by an improvement of the input dataset quality, focusing on detectability, completeness, and finer resolution. First and foremost, increased monitoring densification over the footprint of seismically active regions (Correa-Otto et al., 2018) will be critical for catalogue improvement. In many cases, events in the Neuquén Basin were only recorded by a handful of stations, impacting the detection completeness and spatial resolution of the input catalogue. Substantial improvements can also be realized through the deployment of advanced processing methodologies, such as phase picking via machine learning (Mousavi and Beroza, 2022) or template-matching (Skoumal et al., 2014). Both techniques have shown significant promise for improving catalogue sizes by up to an order-of-magnitude. Deeper catalogues will enable the use of advanced methods that can disentangle competing causal factors (Petrillo et al., 2024; Grigoratos et al., 2022). That said, ensuring that these datasets are open, transparent, and complete will be a key consideration (Schultz et al., 2020). Overall, improving dataset quality will be important for the future of earthquake risk management in the Neuquén Basin (Zhou et al., 2024).

5 Conclusions

In conclusion, we find that many of the earthquakes recorded in the Neuquén Basin are likely induced by hydraulic fracturing. This assertion is based on strong statistical evidence (>99.99% confidence) linking the timings and locations of earthquakes to operations. The lower bound estimate of wells that induced earthquakes is ~0.5%, with maximum magnitudes reaching M4. We further infer causation based on a similarity comparison against known hydraulic fracturing cases around the world. While these analyses indicate events that were induced (~42%), there remains others without clear associations. Whether these events are truly induced by hydraulic fracturing (but lacking an association due to incomplete data), induced by other anthropogenic operations, or part of the natural/tectonic background seismicity is unclear. This lack of clarity could be resolved with additional data (and then further study).

Acknowledgements

We would like to thank the anonymous reviewers whose comments helped improve this work. This work was completed without financial support.

Data and code availability

The data and codes used to produce the results and figures of this study are available online for download at GitHub (https://github.com/RyanJamesSchultz/VM-ID). These resources are also available online at Zenodo (https://doi.org/10.5281/zenodo.14065345). The earthquake catalogue can also be found online at http://contenidos.inpres.gob.ar/buscar_sismo. Well information were derived from the Argentina Energy Secretariat (https://datos.gob.ar/dataset/energia-produccion-petroleo-gas-por-pozo-capitulo-iv/archivo/energia_0abd7bf1-c184-4940-b8cf-5f37961a2ba8) and the Ministry for Energy and Natural Resources of the Neuquén Province

(http://hidrocarburos.energianeuquen.gov.ar/gis).

Competing interests

The authors declare no competing interests.

References

- Ader, T., Chendorain, M., Free, M., Saarno, T., Heikkinen, P., Malin, P. E., Leary, P., Kwiatek, G., Dresen, G., Bluemle, F., and Vuorinen, T. Design and implementation of a traffic light system for deep geothermal well stimulation in Finland. *Journal of Seismology*, 24(5):991–1014, Aug. 2019. doi: 10.1007/s10950-019-09853-y.
- Atkinson, G. M., Eaton, D. W., Ghofrani, H., Walker, D., Cheadle, B., Schultz, R., Shcherbakov, R., Tiampo, K., Gu, J., Harrington, R. M., Liu, Y., van der Baan, M., and Kao, H. Hydraulic Fracturing and Seismicity in the Western Canada Sedimentary Basin. *Seismological Research Letters*, 87(3):631–647, Mar. 2016. doi: 10.1785/0220150263.

- Atkinson, G. M., Eaton, D. W., and Igonin, N. Developments in understanding seismicity triggered by hydraulic fracturing. *Nature Reviews Earth & Environment*, 1(5):264–277, May 2020. doi: 10.1038/s43017-020-0049-7.
- Badessich, M., Hryb, D., Suarez, M., Mosse, L., Palermo, N., Pichon, S., and Reynolds, L. Vaca Muerta shale—Taming a giant. *Oilfield review*, 28(1):26–39, 2016. https://www.slb.com/-/media/files/ oilfield-review/04-vaca-muerta-shale-english.
- Bao, X. and Eaton, D. W. Fault activation by hydraulic fracturing in western Canada. *Science*, 354(6318):1406–1409, Dec. 2016. doi: 10.1126/science.aag2583.
- Beascochea, H. Shell retomó sus trabajos en Vaca Muerta. Vaca Muerta News, 2020. https://vacamuertanews.com/actualidad/ 20200609232301.htm.
- Berger, V. W. and Zhou, Y. Kolmogorov–Smirnov Test: Overview, Sept. 2014. doi: 10.1002/9781118445112.stat06558.
- Bilek, S. L. Invited review paper: Seismicity along the South American subduction zone: Review of large earthquakes, tsunamis, and subduction zone complexity. *Tectonophysics*, 495(1–2): 2–14, Nov. 2010. doi: 10.1016/j.tecto.2009.02.037.
- Bohm, M., Lüth, S., Echtler, H., Asch, G., Bataille, K., Bruhn, C., Rietbrock, A., and Wigger, P. The Southern Andes between 36° and 40°S latitude: seismicity and average seismic velocities. *Tectonophysics*, 356(4):275–289, Oct. 2002. doi: 10.1016/s0040-1951(02)00399-2.
- Bommer, J. J., Oates, S., Cepeda, J. M., Lindholm, C., Bird, J., Torres, R., Marroquín, G., and Rivas, J. Control of hazard due to seismicity induced by a hot fractured rock geothermal project. *Engineering Geology*, 83(4):287–306, Mar. 2006. doi: 10.1016/j.enggeo.2005.11.002.
- Brudzinski, M. R. and Kozłowska, M. Seismicity induced by hydraulic fracturing and wastewater disposal in the Appalachian Basin, USA: a review. *Acta Geophysica*, 67(1):351–364, Jan. 2019. doi: 10.1007/s11600-019-00249-7.
- Cingolani, C., Zanettini, J., and Leanza, H. El basamento ígneo y metamórfico. In Leanza, H., Arregui, C., Carbone, O., Danieli, J., and Vallés, J., editors, *Geología y Recursos Naturales de la Provincia del Neuquén: Relatorio del 18° Congreso Geológico Argentino*, page 37–47. Neuquén, Argentina, 2011.
- Clarke, H., Verdon, J. P., Kettlety, T., Baird, A. F., and Kendall, J. Real-Time Imaging, Forecasting, and Management of Human-Induced Seismicity at Preston New Road, Lancashire, England. Seismological Research Letters, July 2019. doi: 10.1785/0220190110.
- Cobbold, P. and Rossello, E. Aptian to recent compressional deformation, foothills of the Neuquén Basin, Argentina. *Marine and Petroleum Geology*, 20(5):429–443, May 2003. doi: 10.1016/s0264-8172(03)00077-1.
- Corlett, H., Schultz, R., Branscombe, P., Hauck, T., Haug, K., Mac-Cormack, K., and Shipman, T. Subsurface faults inferred from reflection seismic, earthquakes, and sedimentological relationships: Implications for induced seismicity in Alberta, Canada. *Marine and Petroleum Geology*, 93:135–144, May 2018. doi: 10.1016/j.marpetgeo.2018.03.008.

Correa-Otto, S. Experimento sismológico en la cuenca neuquina, la región de mayor explotación de hidrocarburos por métodos no convencionales de la Argentina. Phd thesis, Universidad Nacional de San Juan, 2021. https://www.fundaciongarciasineriz.es/2023/11/20/ experimento-sismologico-en-la-cuenca-neuquina-la-regionde-mayor-explotacion-de-hidrocarburos-por-metodos-noconvencionales-de-la-argentina/.

Correa-Otto, S., Nacif, S., Pesce, A., Nacif, A., Gianni, G., Furlani, R., Giménez, M., and Francisco, R. Intraplate seismicity recorded by a local network in the Neuquén Basin, Argentina. *Journal* of South American Earth Sciences, 87:211–220, Nov. 2018. doi: 10.1016/j.jsames.2017.12.007.

- Correa-Otto, S., Gianni, G., Giménez, M., Spagnotto, S., and Godoy,
 L. Seismotectonic and gravimetric analysis of the central Neuquén Basin. *Journal of South American Earth Sciences*, 144: 105036, Sept. 2024. doi: 10.1016/j.jsames.2024.105036.
- Cristallini, E., Tomezzoli, R., Pando, G., Gazzera, C., Martínez, J., Quiroga, J., and Zambrano, O. Controles precuyanos en la estructura de la CuencaNeuquina. *Revista de la Asociación Geológica Argentina*, 65(2):248–264, 2009.
- Cunningham, N. Fracking in Argentina's Vaca Muerta leads to earthquakes, 2022. https://gasoutlook.com/analysis/frackingin-argentinas-vaca-muerta-leads-to-earthquakes/.
- Curia, D., Duncan, P. M., Grealy, M., McKenna, J., and Hill, A. Microseismic monitoring of Vaca Muerta completions in the Neuquén Basin, Argentina. *The Leading Edge*, 37(4):262–269, Apr. 2018. doi: 10.1190/tle37040262.1.
- Dalmati, R., Souto, J., Zarauz, F., Pertierra, A., Escuder-Bueno, I., and Morales-Torres, A. Implementation of Risk Analysis to Inform Dam Safety from a Regulatory Perspective: Application to Cerros Colorados System (Neuquén, Argentina). In *Twenty-Sixth International Congress on Large Dams/Vingt-Sixième Congrès International des Grands Barrages*, page 101 166– 101 180. CRC Press, 2018.
- Davis, S. D. and Frohlich, C. Did (Or Will) Fluid Injection Cause Earthquakes? - Criteria for a Rational Assessment. Seismological Research Letters, 64(3–4):207–224, July 1993. doi: 10.1785/gssrl.64.3-4.207.
- Eaton, D. W. and Schultz, R. Increased likelihood of induced seismicity in highly overpressured shale formations. *Geophysical Journal International*, 214(1):751–757, May 2018. doi: 10.1093/gji/ggy167.
- Eaton, D. W., Igonin, N., Poulin, A., Weir, R., Zhang, H., Pellegrino, S., and Rodriguez, G. Induced Seismicity Characterization during Hydraulic-Fracture Monitoring with a Shallow-Wellbore Geophone Array and Broadband Sensors. *Seismological Research Letters*, 89(5):1641–1651, July 2018. doi: 10.1785/0220180055.
- Ellsworth, W. L. Injection-Induced Earthquakes. *Science*, 341 (6142), July 2013. doi: 10.1126/science.1225942.
- Farahbod, A. M., Kao, H., Walker, D. M., and Cassidy, J. F. Investigation of regional seismicity before and after hydraulic fracturing in the Horn River Basin, northeast British Columbia. *Canadian Journal of Earth Sciences*, 52(2):112–122, Feb. 2015. doi: 10.1139/cjes-2014-0162.
- Fasola, S. L., Brudzinski, M. R., Skoumal, R. J., Langenkamp, T., Currie, B. S., and Smart, K. J. Hydraulic Fracture Injection Strategy Influences the Probability of Earthquakes in the Eagle Ford Shale Play of South Texas. *Geophysical Research Letters*, 46(22): 12958–12967, Nov. 2019. doi: 10.1029/2019gl085167.
- Forni, L., Escobar, M., Cello, P., Marizza, M., Nadal, G., Girardin, L., Losano, F., Bucciarelli, L., Young, C., and Purkey, D. Navigating the Water-Energy Governance Landscape and Climate Change Adaptation Strategies in the Northern Patagonia Region of Argentina. *Water*, 10(6):794, June 2018. doi: 10.3390/w10060794.
- Forni, L., Mautner, M., Lavado, A., Burke, K., and Gomez, R. *Watershed implications of shale oil and gas production in Vaca Muerta*. SEI Working Paper, Argentina, 2021.
- Foulger, G. R., Wilson, M. P., Gluyas, J. G., Julian, B. R., and Davies, R. J. Global review of human-induced earthquakes. *Earth-Science Reviews*, 178:438–514, Mar. 2018. doi: 10.1016/j.earscirev.2017.07.008.

- Foulger, G. R., Wilkinson, M. W., Wilson, M. P., Mhana, N., Tezel, T., and Gluyas, J. G. Human-induced earthquakes: E-PIE—a generic tool for Evaluating Proposals of Induced Earthquakes. *Journal of Seismology*, 27(1):21–44, Jan. 2023. doi: 10.1007/s10950-022-10122-8.
- Frohlich, C., DeShon, H., Stump, B., Hayward, C., Hornbach, M., and Walter, J. I. A Historical Review of Induced Earthquakes in Texas. *Seismological Research Letters*, 87(4):1022–1038, May 2016. doi: 10.1785/0220160016.
- Galloway, E., Hauck, T., Corlett, H., Pană, D., and Schultz, R. Faults and associated karst collapse suggest conduits for fluid flow that influence hydraulic fracturing-induced seismicity. *Proceedings of the National Academy of Sciences*, 115(43), Oct. 2018. doi: 10.1073/pnas.1807549115.
- Ghofrani, H. and Atkinson, G. Reply to "Comment on 'Activation Rate of Seismicity for Hydraulic Fracture Wells in the Western Canadian Sedimentary Basin'by Hadi Ghofrani and Gail M. *Bulletin of the Seismological Society of America*, 111(6):3475–3497, 2021. doi: 10.1785/0120210059.
- Ghofrani, H. and Atkinson, G. M. Activation Rate of Seismicity for Hydraulic Fracture Wells in the Western Canada Sedimentary Basin. *Bulletin of the Seismological Society of America*, 110(5): 2252–2271, May 2020. doi: 10.1785/012020002.
- Grigoratos, I., Savvaidis, A., and Rathje, E. Distinguishing the Causal Factors of Induced Seismicity in the Delaware Basin: Hydraulic Fracturing or Wastewater Disposal? *Seismological Research Letters*, 93(5):2640–2658, June 2022. doi: 10.1785/0220210320.
- Gupta, H. K. A review of recent studies of triggered earthquakes by artificial water reservoirs with special emphasis on earthquakes in Koyna, India. *Earth-Science Reviews*, 58(3–4):279–310, Oct. 2002. doi: 10.1016/s0012-8252(02)00063-6.
- Healy, J. H., Rubey, W. W., Griggs, D. T., and Raleigh, C. B. The Denver Earthquakes. *Science*, 161(3848):1301–1310, Sept. 1968. doi: 10.1126/science.161.3848.1301.
- Hicks, S. P., Goes, S., Whittaker, A. C., and Stafford, P. J. Multivariate Statistical Appraisal of Regional Susceptibility to Induced Seismicity: Application to the Permian Basin, SW United States. *Journal of Geophysical Research: Solid Earth*, 126(12), Dec. 2021. doi: 10.1029/2021jb022768.
- Horton, S. Disposal of Hydrofracking Waste Fluid by Injection into Subsurface Aquifers Triggers Earthquake Swarm in Central Arkansas with Potential for Damaging Earthquake. *Seismological Research Letters*, 83(2):250–260, Mar. 2012. doi: 10.1785/gssrl.83.2.250.
- Howell, J. A., Schwarz, E., Spalletti, L. A., and Veiga, G. D. The Neuquén Basin: an overview. *Geological Society, London, Special Publications*, 252(1):1–14, Jan. 2005. doi: 10.1144/gsl.sp.2005.252.01.01.
- Häring, M. O., Schanz, U., Ladner, F., and Dyer, B. C. Characterisation of the Basel 1 enhanced geothermal system. *Geothermics*, 37(5):469–495, Oct. 2008. doi: 10.1016/j.geothermics.2008.06.002.
- Jacquemond, L., Letort, J., Cotton, F., Causse, M., Grasso, J. R., Senechal, G., Ammirati, J. B., Derode, B., Grimaud, F., Pauchet, H., Benhamed, S., and Sylvander, M. Analysing 50 yr of the Lacq induced seismicity (Southwestern, France) highlights the role of fluid injection. *Geophysical Journal International*, 238(1): 214–234, Mar. 2024. doi: 10.1093/gji/ggae119.
- Kettlety, T., Verdon, J. P., Butcher, A., Hampson, M., and Craddock, L. High-Resolution Imaging of the ML 2.9 August 2019 Earthquake in Lancashire, United Kingdom, Induced by Hydraulic Fracturing during Preston New Road PNR-2 Operations. *Seismological Research Letters*, 92(1):151–169, Oct. 2020. doi:

10.1785/0220200187.

- Kugler, R. L. Source rock characteristics, Los Molles and Vaca Muerta shales, Neuquén Basin, west-central Argentina. AAPG Bulletin, 69, 1985. doi: 10.1306/ad4620ce-16f7-11d7-8645000102c1865d.
- Legarreta, L. and Villar, H. The Vaca Muerta Formation (Late Jurassic-Early Cretaceous). In *Unconventional Resources Technology Conference*, Neuquen Basin, Argentina, 2015. sequences. doi: 10.15530/urtec-2015-2170906.
- Lei, X., Huang, D., Su, J., Jiang, G., Wang, X., Wang, H., Guo, X., and Fu, H. Fault reactivation and earthquakes with magnitudes of up to Mw4.7 induced by shale-gas hydraulic fracturing in Sichuan Basin, China. *Scientific Reports*, 7(1), Aug. 2017. doi: 10.1038/s41598-017-08557-y.
- Lei, X., Wang, Z., and Su, J. The December 2018 ML 5.7 and January 2019 ML 5.3 Earthquakes in South Sichuan Basin Induced by Shale Gas Hydraulic Fracturing. *Seismological Research Letters*, 90(3):1099–1110, Apr. 2019. doi: 10.1785/0220190029.
- LM Neuquen. Se entregaron viviendas antisísmicas para vecinos de Sauzal, 2024. https://www.lmneuquen.com/neuquen/ se-entregaron-viviendas-antisismicas-vecinos-sauzal-bonito- n1104994.
- Lomax, A. and Savvaidis, A. Improving Absolute Earthquake Location in West Texas Using Probabilistic, Proxy Ground-Truth Station Corrections. *Journal of Geophysical Research: Solid Earth*, 124(11):11447–11465, Nov. 2019. doi: 10.1029/2019jb017727.
- Lu, X., Li, S., Han, R., Jia, B., Xian, C., Ding, F., Zhang, Y., Zhong, S., Xu, Q., and Zuo, X. An Overview of Recent Developments and Understandings of Unconventionals in the Vaca Muerta Formation, Argentina. *Applied Sciences*, 14(4):1366, Feb. 2024. doi: 10.3390/app14041366.
- Mahani, A. B., Schultz, R., Kao, H., Walker, D., Johnson, J., and Salas, C. Fluid Injection and Seismic Activity in the Northern Montney Play, British Columbia, Canada, with Special Reference to the 17 August 2015Mw 4.6 Induced Earthquake. *Bulletin* of the Seismological Society of America, 107(2):542–552, Feb. 2017. doi: 10.1785/0120160175.
- Marchal, D., Manceda, R., Domínguez, R., and Sattler, F. Structural geology: Tectonic history, macrostructures, regional fault map, fault systems, second-order structures, and impact of the inheritance. In Minisini, D., Fantín, M., Noguera, I. L., and Leanza, H. A., editors, *Integrated geology of unconventionals: The case of the Vaca Muerta play*, volume 121, page 99–140. AAPG Memoir, Argentina, 2020. doi: 10.1306/13682225M1203831.
- Martínez, M. A., Prámparo, M. B., Quattrocchio, M. E., and Zavala, C. A. Depositional environments and hydrocarbon potential of the Middle Jurassic Los Molles Formation, Neuquén Basin, Argentina: palynofacies and organic geochemical data. *Andean Geology*, 35(2), June 2008. doi: 10.5027/andgeov35n2-a05.
- Minisini, D., Fryklund, B., Gerali, F., and Fantín, M. The first economical unconventional play outside North America: Context, history, and "coopetition". In Minisini, D., Fantín, M., Noguera, I. L., and Leanza, H. A., editors, *Integrated geology* of unconventionals: The case of the Vaca Muerta play, volume 121, page 1–24. AAPG Memoir, Argentina, 2020. doi: 10.1306/13682221M1203264.
- Moein, M. J. A., Langenbruch, C., Schultz, R., Grigoli, F., Ellsworth, W. L., Wang, R., Rinaldi, A. P., and Shapiro, S. The physical mechanisms of induced earthquakes. *Nature Reviews Earth & Environment*, 4(12):847–863, Dec. 2023. doi: 10.1038/s43017-023-00497-8.
- Mosquera, A., Alonso, J., Boll, A., Alarcón, M., Zavala, C., Arcuri, M., and Villar, H. Migración lateral y evidencias de hidrocarburos cuyanos en yacimientos de la plataforma de Catriel, Cuenca

Neuquina. In 7 Congreso de Exploración y Desarrollo de Hidrocarburos, Mar del Plata, Argentina, page 5–8. IAPG, Buenos Aires, 2008.

- Mousavi, S. M. and Beroza, G. C. Deep-learning seismology. *Science*, 377(6607), Aug. 2022. doi: 10.1126/science.abm4470.
- Muntendam-Bos, A. G., Hoedeman, G., Polychronopoulou, K., Draganov, D., Weemstra, C., van der Zee, W., Bakker, R. R., and Roest, H. An overview of induced seismicity in the Netherlands. *Netherlands Journal of Geosciences*, 101, 2022. doi: 10.1017/njg.2021.14.
- Nacion. Neuquén: un hombre murió aplastado en el derrumbe de un acantilado en el lago Mari Menuco, La Nacion, 2022. https://www.lanacion.com.ar/seguridad/neuquen-unhombre-murio-aplastado-en-el-derrumbe-de-un-acantiladoen-el-lago-mari-menuco-nid10022022/.
- Oprsal, I. and Eisner, L. Cross-correlation—an objective tool to indicate induced seismicity. *Geophysical Journal International*, 196(3):1536–1543, Jan. 2014. doi: 10.1093/gji/ggt501.
- Pawley, S., Schultz, R., Playter, T., Corlett, H., Shipman, T., Lyster, S., and Hauck, T. The Geological Susceptibility of Induced Earthquakes in the Duvernay Play. *Geophysical Research Letters*, 45 (4):1786–1793, Feb. 2018. doi: 10.1002/2017gl076100.
- Petrillo, G., Kumazawa, T., Napolitano, F., Capuano, P., and Zhuang, J. Fluids-Triggered Swarm Sequence Supported by a Nonstationary Epidemic-Like Description of Seismicity. *Seismological Research Letters*, 95(6):3207–3220, June 2024. doi: 10.1785/0220240056.
- Peña Castro, A. F., Roth, M. P., Verdecchia, A., Onwuemeka, J., Liu, Y., Harrington, R. M., Zhang, Y., and Kao, H. Stress Chatter via Fluid Flow and Fault Slip in a Hydraulic Fracturing-Induced Earthquake Sequence in the Montney Formation, British Columbia. *Geophysical Research Letters*, 47(14), July 2020. doi: 10.1029/2020gl087254.
- Raleigh, C. B., Healy, J. H., and Bredehoeft, J. D. An Experiment in Earthquake Control at Rangely, Colorado. *Science*, 191(4233): 1230–1237, Mar. 1976. doi: 10.1126/science.191.4233.1230.
- Ramos, V., Naipauer, M., Leanza, H., and Sigismondi, M. An exceptional tectonic setting along the Andean continental margin.
 In Minisini, D., Fantín, M., Noguera, I. L., and Leanza, H. A., editors, *Integrated geology of unconventionals: The case of the Vaca Muerta play*, volume 121, page 25–37. AAPG Memoir, Argentina, 2020. doi: 10.1306/13682222M1202855.
- Rosa, L. and D'Odorico, P. The water-energy-food nexus of unconventional oil and gas extraction in the Vaca Muerta Play, Argentina. *Journal of Cleaner Production*, 207:743–750, Jan. 2019. doi: 10.1016/j.jclepro.2018.10.039.
- Savvaidis, A., Lomax, A., and Breton, C. Induced Seismicity in the Delaware Basin, West Texas, is Caused by Hydraulic Fracturing and Wastewater Disposal. *Bulletin of the Seismological Society of America*, 110(5):2225–2241, Aug. 2020. doi: 10.1785/0120200087.
- Schultz, R. and Telesca, L. The Cross-Correlation and Reshuffling Tests in Discerning Induced Seismicity. *Pure and Applied Geophysics*, 175(10):3395–3401, May 2018. doi: 10.1007/s00024-018-1890-1.
- Schultz, R. and Wang, R. Newly emerging cases of hydraulic fracturing induced seismicity in the Duvernay East Shale Basin. *Tectonophysics*, 779:228393, Mar. 2020. doi: 10.1016/j.tecto.2020.228393.
- Schultz, R., Wang, R., Gu, Y. J., Haug, K., and Atkinson, G. A seismological overview of the induced earthquakes in the Duvernay play near Fox Creek, Alberta. *Journal of Geophysical Research: Solid Earth*, 122(1):492–505, Jan. 2017. doi: 10.1002/2016jb013570.

- Schultz, R., Atkinson, G., Eaton, D. W., Gu, Y. J., and Kao, H. Hydraulic fracturing volume is associated with induced earthquake productivity in the Duvernay play. *Science*, 359(6373): 304–308, Jan. 2018. doi: 10.1126/science.aa00159.
- Schultz, R., Skoumal, R. J., Brudzinski, M. R., Eaton, D., Baptie, B., and Ellsworth, W. Hydraulic Fracturing-Induced Seismicity. *Reviews of Geophysics*, 58(3), July 2020. doi: 10.1029/2019rg000695.
- Schultz, R., Beroza, G. C., and Ellsworth, W. L. A riskbased approach for managing hydraulic fracturing-induced seismicity. *Science*, 372(6541):504–507, Apr. 2021a. doi: 10.1126/science.abg5451.
- Schultz, R., Beroza, G. C., and Ellsworth, W. L. A Strategy for Choosing Red-Light Thresholds to Manage Hydraulic Fracturing Induced Seismicity in North America. *Journal of Geophysical Research: Solid Earth*, 126(12), Dec. 2021b. doi: 10.1029/2021jb022340.
- Schultz, R., Ellsworth, W. L., and Beroza, G. C. Statistical bounds on how induced seismicity stops. *Scientific Reports*, 12(1), Jan. 2022. doi: 10.1038/s41598-022-05216-9.
- Schultz, R., Baptie, B., Edwards, B., and Wiemer, S. Red-light thresholds for induced seismicity in the UK. *Seismica*, 2(2), Oct. 2023a. doi: 10.26443/seismica.v2i2.1086.
- Schultz, R., Ellsworth, W. L., and Beroza, G. C. An Ensemble Approach to Characterizing Trailing-Induced Seismicity. *Seismological Research Letters*, 94(2A):699–707, Jan. 2023b. doi: 10.1785/0220220352.
- Schultz, R., Park, Y., Aguilar Suarez, A. L., Ellsworth, W. L., and Beroza, G. C. En echelon faults reactivated by wastewater disposal near Musreau Lake, Alberta. *Geophysical Journal International*, 235(1):417–429, May 2023c. doi: 10.1093/gji/ggad226.
- Silvestro, J. and Zubiri, M. Convergencia oblicua: modelo estructural alternativo para la Dorsal Neuquina (39oS)-Neuquén. *Revista de la Asociación Geológica Argentina*, 63(1):49–64, 2008.
- Skoumal, R. J., Brudzinski, M. R., Currie, B. S., and Levy, J. Optimizing multi-station earthquake template matching through re-examination of the Youngstown, Ohio, sequence. *Earth* and Planetary Science Letters, 405:274–280, Nov. 2014. doi: 10.1016/j.epsl.2014.08.033.
- Skoumal, R. J., Brudzinski, M. R., and Currie, B. S. Earthquakes Induced by Hydraulic Fracturing in Poland Township, Ohio. *Bulletin of the Seismological Society of America*, 105(1):189–197, Jan. 2015. doi: 10.1785/0120140168.
- Skoumal, R. J., Brudzinski, M. R., and Currie, B. S. Proximity of Precambrian basement affects the likelihood of induced seismicity in the Appalachian, Illinois, and Williston Basins, central and eastern United States. *Geosphere*, 14(3):1365–1379, Apr. 2018. doi: 10.1130/ges01542.1.
- Skoumal, R. J., Barbour, A. J., Brudzinski, M. R., Langenkamp, T., and Kaven, J. O. Induced Seismicity in the Delaware Basin, Texas. *Journal of Geophysical Research: Solid Earth*, 125(1), Jan. 2020. doi: 10.1029/2019jb018558.
- Surma, K. Fracking-induced earthquakes are menacing Argentina as regulators stand by. *Inside Climate News*, 2024. https://insideclimatenews.org/news/14042024/argentinafracking-earthquakes/.
- Tamburini-Beliveau, G., Grosso-Heredia, J. A., Béjar-Pizarro, M., Pérez-López, R., Portela, J., Cismondi-Duarte, M., and Monserrat, O. Assessment of ground deformation and seismicity in two areas of intense hydrocarbon production in the Argentinian Patagonia. *Scientific Reports*, 12(1), Nov. 2022. doi: 10.1038/s41598-022-23160-6.
- Urien, C. and Zambrano, J. Petroleum Systems in the Neuquen

Basin, Argentina: Chapter 32: Part V Case Studies–Western Hemisphere, 1994. doi: 10.1306/M60585C32.

- US-EIA, U. I. A. Technically Recoverable Shale Oil and Shale Gas Resources: An Assessment of 137 Shale Formations in 41 Countries Outside the United States. In US Department of Energy. 2013. https://www.eia.gov/analysis/studies/worldshalegas/ pdf/overview.pdf.
- US-EIA, U. I. A. Technically recoverable shale oil and shale gas resources: Argentina. In US Department of Energy. 2015. https: //www.eia.gov/analysis/studies/worldshalegas/.
- Varela, R., Marchal, D., Cuervo, S., Lombardo, E., Perl, Y., Hryb, D., and Pateti, P. Geomechanics: Pressure, stress field, and hydraulic fractures. In Minisini, D., Fantín, M., Noguera, I. L., and Leanza, H. A., editors, *Integrated geology of unconventionals: The case of the Vaca Muerta play*, volume 121, page 351–376. AAPG Memoir, Argentina, 2020. doi: 10.1306/13682233M1203836.
- Verdon, J. P. and Bommer, J. J. Comment on "Activation Rate of Seismicity for Hydraulic Fracture Wells in the Western Canadian Sedimentary Basin" by Hadi Ghofrani and Gail M. Atkinson. *Bulletin of the Seismological Society of America*, 111(6):3459–3474, Oct. 2021. doi: 10.1785/0120200350.
- Verdon, J. P., Baptie, B. J., and Bommer, J. J. An Improved Framework for Discriminating Seismicity Induced by Industrial Activities from Natural Earthquakes. *Seismological Research Letters*, Apr. 2019. doi: 10.1785/0220190030.
- Vásquez, J., Spagnotto, S., Mescua, J., Giambiagi, L., and Sigismondi, M. Aumento notorio de la sismicidad de la provincia del Neuquén, en el período 2015-2020. In *Boletín Brackebuschiano Geociencias y Sociedad de la Asociación Geológica Argentina*. 2020. https://geologica.org.ar/wp-content/uploads/2020/12/ boletin-octubre-noviembre-2020.pdf.
- Wang, J., Li, T., Gu, Y. J., Schultz, R., Yusifbayov, J., and Zhang, M. Sequential Fault Reactivation and Secondary Triggering in the March 2019 Red Deer Induced Earthquake Swarm. *Geophysical Research Letters*, 47(22), Nov. 2020. doi: 10.1029/2020gl090219.
- Wetmiller, R. J. Earthquakes near Rocky Mountain House, Alberta, and their relationship to gas production facilities. *Canadian Journal of Earth Sciences*, 23(2):172–181, Feb. 1986. doi: 10.1139/e86-020.
- Zhai, G., Shirzaei, M., and Manga, M. Widespread deep seismicity in the Delaware Basin, Texas, is mainly driven by shallow wastewater injection. *Proceedings of the National Academy of Sciences*, 118(20), May 2021. doi: 10.1073/pnas.2102338118.
- Zhou, W., Lanza, F., Grigoratos, I., Schultz, R., Cousse, J., Trutnevyte, E., Muntendam-Bos, A., and Wiemer, S. Managing Induced Seismicity Risks From Enhanced Geothermal Systems: A Good Practice Guideline. *Reviews of Geophysics*, 62(4), Oct. 2024. doi: 10.1029/2024rg000849.
- Álvarez Mullally, M. Vaca Muerta: sesenta sismos en diez días, OPSur, 2021. https://opsur.org.ar/2021/07/27/vaca-muertasesenta-sismos-en-diez-dias/.

The article Chasing the ghost of fracking in the Vaca Muerta Formation: Induced seismicity in the Neuquén Basin, Argentina © 2024 by Ryan Schultz is licensed under CC BY 4.0.